Author’s response to reviews

Title: Men’s grief following pregnancy loss and neonatal loss: A systematic review and emerging theoretical model

Authors:

Kate Obst (kate.obst@adelaide.edu.au)
Clemence Due (clemence.due@adelaide.edu.au)
Melissa Oxlad (melissa.oxlad@adelaide.edu.au)
Philippa Middleton (philippa.middleton@adelaide.edu.au)

Version: 3 Date: 09 Dec 2019

Author’s response to reviews:

Dear Lauren and Janelle,

Thank you for your feedback on our manuscript, "Men’s grief following pregnancy loss and neonatal loss: A systematic review and emerging theoretical model" (PRCH-D-19-01275R1).

Please find below our responses to your comments, along with the final, clean version of our manuscript, and associated files, attached. I had emailed to confirm that the changes made were adequate, however did not receive a response in time for the submission deadline. As such, I do hope these changes are sufficient.

Thank you again for your consideration. We look forward to hearing from you soon regarding the outcome.

Kindest regards,
Ms Kate Obst
Dr Clemence Due
Dr Melissa Oxlad and
A/Prof Philippa Middleton

--

1. Thank you for clarifying the contributions of MO and PM, however, we would like to request that you please include the contributions of MO and PM in the Authors Contributions section.

- The Authors Contribution section has been updated to include the contributions of MO and PM.

2. *As a reminder*, please proofread and ensure that when you upload your revised submission it is in the final form for publication. Please remove any tracked changes or highlighting and
include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

- All highlighting, tracked changes and line numbers have been removed. The revised submission has been uploaded in its final form for publication as a single clean copy.